Author's response to reviews

Title: Blood transfusion presenting with acute dyspnoea: a case report and review of literature

Authors:

SHEKHAR SHARMA (drshekharsharma@gmail.com)
ALTAF GAUHAR (altafhaji786@yahoo.com)
VIJAYKUMAR D K (dkvijaykumar@aims.amrita.edu)
JERRY PAUL (jerrypaul@aims.amrita.edu)

Version: 2 Date: 14 July 2008

Author's response to reviews:

The revision is in conformity with the suggestions made by the editorial team vide their e-mail dated March 14th 2008